Author's response to reviews

Title: Systematic review of healthcare interventions to improve outcomes for women with disability and their family during pregnancy, birth and postnatal period

Authors:

Reem Malouf (reem.malouf@npeu.ox.ac.uk)
Maggie Redshaw (maggie.redshaw@npeu.ox.ac.uk)
Jennifer Kurinczuk (jenny.kurinczuk@npeu.ox.ac.uk)
Ron Gray (ron.gray@npeu.ox.ac.uk)

Version: 2 Date: 16 January 2014

Author's response to reviews:

Dr Edwin van Teijlingen
Editor
BMC Pregnancy and Childbirth
Re: MS 1624075980991053

Systematic review of healthcare interventions to improve outcomes for women with disability and their families during pregnancy, birth and postnatal period
Reem Malouf, Maggie Redshaw, Jennifer J Kurinczuk and Ron Gray BMC Pregnancy and Childbirth (Section: Maternity care and sociological aspects of pregnancy and childbirth)

Dear Dr van Teijlingen,

Thank you for the opportunity to resubmit this paper with revisions. We have responded to the referees' comments (marked in red) and have made additional changes to the manuscript (marked in yellow). In particular we have taken this opportunity to try and clarify our aims and methods.

I hope our revised manuscript now meets the standard for publication in BMC Pregnancy and Childbirth and we look forward to hearing from you soon.

Best Wishes,

Reem Malouf